Author's response to reviews

Title: Factors associated with mortality in HIV-infected and uninfected patients with pulmonary tuberculosis

Authors:

Ferdinand M Mugusi (fmugusi@muhas.ac.tz)
Saurabh Mehta (smehta@hsph.harvard.edu)
Eduardo Villamor (evillamor@hsph.harvard.edu)
Willy Urassa (wurassa@muhas.ac.tz)
Elma Saathoff (esaathoff@muchs-harvard.org)
Ronald Bosch (rbosch@hsph.harvard.edu)
Wafaie Fawzi (mina@hsph.harvard.edu)

Version: 3 Date: 23 October 2009

Author's response to reviews:

Dear Editor,

Thank you for accepting our manuscript for publication in your esteemed journal. The following are the revisions that have been done following your suggestions.

1) The title has been revised to correspond with the title in the submission system
2) Affiliations have been revised to include departments and institutions of collaborating authors.
3) The abstract is identical in the manuscript and the submission system.
4) All authors have read and approved the submission of the manuscript.
5) All tables have been revised without changing the contents to fit the required size. Table 2 which was initially 5 pages long is currently 3 pages long. The rest are less than 3 pages long.
6) All titles of tables are above the table and legends are below the table.
7) The manuscript has been checked for typographical errors as far as possible.

It is my sincere hope that this will meet your satisfaction.

Ferdinand Mugusi